Author’s response to reviews

Title: SNHG 6 Promotes the Progression of Colon and Rectal Adenocarcinoma via miR-101-3p and Wnt/β-catenin Signaling Pathway

Authors:

Qianwen Shao (zqw1990@tom.com)
Jing Xu (xujing83@foxmail.com)
Rong Deng (dengrongrx@163.com)
Wei Wei (what217@163.com)
Bing Zhou (zbgastric@126.com)
Chao Yue (yuechaopwk@yeah.net)
Miaoling Zhu (zhumiaolin829@126.com)
Haitao Zhu (co87r5@163.com)

Version: 1 Date: 26 Aug 2019

Author’s response to reviews:

Dear Editor and reviewers:

Thank you for your comments concerning our manuscript entitled “SNHG 6 Promotes the Progression of Colon and Rectal Adenocarcinoma via miR-101-3p and Wnt/β-catenin Signaling Pathway” (BMGE-D-19-00619). Those comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the comments are as flowing:

Responds to the comments:

Xiaofei Sun (Reviewer 1 - MINOR REVISIONS):

The whole manuscript need proof reading. For example, delete "via" in line 28 in Abstract section.
Response: We are sorry for our negligence, and we have deleted the word "via" in line 28 in Abstract section. Moreover, we have proofed this manuscript very carefully.

2. The title "SNHG 6 Promotes the Progression of Colon and Rectal Adenocarcinoma via miR-101-3p and Wnt/β-catenin Signaling Pathway A concise and informative SNHG6 in human Colon and Rectal Adenocarcinoma" is too long for readers. Consider to use "SNHG 6 Promotes the Progression of Colon and Rectal Adenocarcinoma via miR-101-3p and Wnt/β-catenin Signaling Pathway".

Response: The title of our manuscript is "SNHG 6 Promotes the Progression of Colon and Rectal Adenocarcinoma via miR-101-3p and Wnt/β-catenin Signaling Pathway". The short title is “SNHG6 in human Colon and Rectal Adenocarcinoma”. Considering the reviewer’s suggestion, we made a more obvious distinction by paragraphing.

Yuusaku Sugihara (Reviewer 2 - ACCEPT): Comments to the Author

In this manuscript, the authors conducted small nucleolar RNA host gene 6 (SNHG6) regulates diverse biological processes in cancers. Potential function of SNHG6 in human colon and rectal adenocarcinoma (CRC) was evaluated.

The conclusion was that SNHG6 could regulate the progression of CRC via via modulating the expression levels of miR-101-3p and the activity of Wnt/β-catenin signaling.

This study represents an effort to accomplish this, as these data will most likely prove to be a unique resource well into the future.

Response: Thank you very much for your comments.

We tried our best to improve the manuscript and list the changes here. We appreciate for Editor’s warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.